     The intensity of a beam propagating through a linearly attenuating medium (for simplicity we consider pure absorption; $C_{attn} \equiv C_{abs}$) will decrease exponentially as given by the Beer-Lambert law (Fig. \ref{fig:overview}(a,b)). For propagation-invariant beams, there is a mapping between the transverse coordinate in the pupil plane and the longitudinal coordinate near the focal plane\cite{Cizmar2009,ZamboniRached2004,Preciado2012,Preciado2014,Schley2014}. Therefore, by judicious choice of weighted complex amplitudes for different wave vectors of the beam, the intensity profile can be engineered such that the net attenuation is zero (Fig. \ref{fig:overview}(c)). The cylindrical pupil function for an attenuation-compensated Airy light-sheet is given by:
    
    \begin{equation}
    \label{eq:MainTextAiryPupil1D}
    	P(u) = A_{\sigma} \exp(2\pi i \alpha u^{3}) \exp(-u^{8}) \exp(\sigma[u-1])H(\sqrt{2}-\abs{u})
    \end{equation}
    
    \noindent where $u$ is the normalised pupil coordinate corresponding to the $z-$axis of the microscope, $A_{\sigma}$ is a real scaling factor, $\alpha$ dictates the propagation-invariance of the Airy light-sheet \cite{Vettenburg2014}, $\sigma$ dictates the degree of linear attenuation-compensation, $H(\cdot)$ denotes the Heaviside step function, and the light-sheet propagates in the positive $x-$direction (see Supplementary Note \ref{supp:AiryTheory}). Similar mappings exist for Bessel beams (see Supplementary Note \ref{supp:BesselTheory}).
    
    When attenuation-compensation is applied to an Airy light-sheet, the effective attenuation coefficient, $C_{attn}'$, then becomes:
    
    \begin{equation}
    \label{eq:MainTextModifieldAttenuationCoeff}
    	\begin{aligned}
    		C_{attn}' &= C_{attn} - \chi \\
            &= C_{attn} - \frac{10 \sigma \NA^{2}}{3 \pi^{2} n \alpha \lambda}
        \end{aligned}
    \end{equation}
    
    \noindent where $\NA$ is the numerical aperture corresponding to $u=1$, $n$ is the refractive index of the sample, and $\lambda$ is the vacuum wavelength of the illumination (see Supplementary Note \ref{supp:AiryTheory})\cite{Preciado2014}. For $\chi = C_{attn}$, the main caustic of the Airy light-sheet propagates as if without any loss.
    
    \begin{figure}[t]
      \centering
      \includegraphics[width=\linewidth]{OverviewFigure.pdf}
      \caption{Ray optics representations of Airy light-sheet formation without (a) and with (b) attenuation ($C_{attn} = 64.95$cm$^{-1}$), and with attenuation-compensation (c; $\sigma = 0.54$). (d-f) Wave optical simulations of light-sheet profiles, (g-i) peak transverse intensity as a function of longitudinal coordinate, (j-o) simulated recorded and deconvolved images of the University crest, and (p-r) simulated deconvolved images of a 1D resolution target respectively for the light-sheets shown in (a-c). Pink dashed lines in (m-o) indicate the edge of the FOV from theory. 1D resolution target (p-r) has line width/spacing: $2\mum$ (top), $1\mum$, $0.6\mum$, $0.2\mum$ (bottom). Simulation parameters were set to mirror experimental parameters (see Methods (Section \ref{sec:methods})). These datasets can be accessed at\cite{Data}.}
      \label{fig:overview}
	\end{figure}
    
    The selective delivery of additional power to greater depths within an attenuating medium allows the Airy light-sheet to maintain consistent intensity along its main caustic (Fig. \ref{fig:overview}(d-i)). However, with our experimental parameters (see Methods (Section \ref{sec:methods})), for $\sigma > 0.54$ we found the peak intensity was no longer located on the main caustic and gave unpredictable longitudinal intensity profiles (see Supplementary Note \ref{supp:AiryTheory}). Therefore we defined $\sigma = 0.54$ as the maximum permitted attenuation-compensation for this study, allowing complete compensation up to $C_{attn} = 64.95$cm$^{-1}$ over the whole FOV of the light-sheet as determined from theory ($328\mum$)\cite{Vettenburg2014}.
     
     Airy LSM utilises deconvolution of the recorded images, based on the point spread function (PSF) of the illuminating light-sheet to achieve high axial resolution. Modification of the deconvolution procedure to incorporate the attenuated light-sheet profile (see Supplementary Note \ref{supp:Deconvolution}) facilitates correct re-normalisation of the sample fluorophore distribution, but at the cost of amplified noise in regions where the illumination is weak. Due to the increased intensity at depth when using attenuation-compensation, the signal is restored without noise amplification (Fig. \ref{fig:overview}(j-r), Supplementary Fig. \ref{fig:LineTarget_images} and \ref{fig:LineTarget_plots}).
     
     As deconvolution is required for Airy LSM, the axial image quality can be explored through analysis of the modulation transfer function (MTF) of the light-sheet profiles. We found that attenuation results in a marked reduction in the support of high spatial frequencies that is restored with the use of attenuation-compensation (see Supplementary Note \ref{supp:AiryImageTheory}). Interestingly, for complete attenuation-compensation ($C_{attn}'=0$) the bandwidth was found to be higher than for the ideal (non-attenuating) case. Deconvolution requires \textit{a priori} knowledge of the sample attenuation and incorrect estimation of $C_{attn}$ will result in image artefacts. Estimation of $C_{attn}$ to within $10\%$ of the true value was sufficient to minimise the effect of such artefacts (see Supplementary Note \ref{supp:DeconvolutionAttnMismatch}).
     
     Similarly, for Bessel LSM methods, the use of attenuation-compensation does not affect the transverse beam shape (see Supplementary Note \ref{supp:BesselImageTheory}) and is therefore compatible will all existing Bessel LSM techniques (e.g. \cite{Fahrbach2010,Fahrbach2012,Zhang2014,Zhang2016,Planchon2011,Chen2014,Gao2012,Olarte2012}).